Author's response to reviews

Title: Severe cholestatic jaundice after a single administration of ajmaline; a case report and review of the literature.

Authors:

Benjamin H Mullish (b.mullish@imperial.ac.uk)
Rishi K Fofaria (rishi.fofaria@nhs.net)
Belinda C Smith (belinda.smith2@imperial.nhs.uk)
Kirsty Lloyd (k.lloyd@imperial.ac.uk)
Josephine Lloyd (j.lloyd@imperial.ac.uk)
Robert D Goldin (r.goldin@imperial.ac.uk)
Ameet Dhar (a.dhar@imperial.ac.uk)

Version: 8 Date: 12 March 2014

Author's response to reviews: see over
Dear Colleague

Re: MS: 311632891129516
Severe cholestatic jaundice after a single administration of ajmaline; a case report and review of the literature.

We were very pleased to receive your positive feedback on this submission from us, and the constructive point suggested by yourself and Reviewer 2 regarding a change to the references. We are happy to resubmit a revised manuscript.

As per the recommendation of Reviewer 2, we have changed the previous Reference 11 for a new Reference 11, 12 and 13, citing the papers that he suggests. The rest of the manuscript is largely unchanged, apart the very occasional minor typographical revision.

We look forward to hearing back from you. Please do not hesitate to get in contact with me via b.mullish@imperial.ac.uk or via the contact details above should you require any further information.

Kind regards

Yours Faithfully

Dr Benjamin Mullish MB BChir MA (Cantab) MRCP (UK)
Academic Clinical Fellow, Hepatology and Gastroenterology
Imperial College London